Author's response to reviews

Title: Rapidly progressive form of Bronchiolitis Obliterans Organising Pneumonia presenting with pneumothorax, persistent and recurrent air-leak, acute respiratory distress syndrome and multi-organ dysfunction

Authors:

Indranil Chakravorty (i.chakravorty@herts.ac.uk)
William LG Oldfield (William.Oldfield@St-Marys.nhs.uk)
Carlos MH Gomez (Carlos.Gomez@St-Marys.nhs.uk)

Version: 2 Date: 21 June 2007

Author's response to reviews:

Dear Clare,

Many thanks for your comments. I have made the necessary changes recommended to make this manuscript conform to the requirements of your journal.

1. The title has been changed to finish with ; a case report
2. The abstract has been shifted to page 2
3. An introduction section has been added
4. A conclusion section has been added
5. References have been reduced to 15 max
6. Acknowledgement, contribution list and consent statements have been added

I hope this interesting and rare case is accepted for publication as this would help to highlight the salient features to a much wider audience and benefit patients.